Author's response to reviews

Title: Antitumor effect of Kanglaite injection in human pancreatic cancer xenografts

Authors:

Ying Liu (liuying2199@yahoo.com.cn)
Xiao-Jie Wang (rps2199@sina.cn)
Wei Zhang (rdning@163.com)
Shi Liu (shiliu2199@163.com)

Version: 2 Date: 3 June 2013

Author's response to reviews: see over
Dear James,

Thank you for your letter and for the comments concerning our manuscript entitled “Antitumor Effect of Kanglaite® Injection in Human Pancreatic Cancer Xenografts” (MS: 8414646901006149). Those comments are all valuable and very helpful for revising and improving our paper. We have studied comments carefully and have made correction which we hope meet with approval. Revised portion are marked in yellow in the paper. The corrections in the paper are as following:

1. We revised our paper conforming to the journal style.
2. In the present study, all animal experiment protocols were approved by Animal Use and Management Committee of Qiqihar Medical University. Nevertheless, the committee does not have any of approval numbers for all the animal experiments in our University.
3. We changed the order of authors.

We appreciate for your warm work earnestly and thank you very much for your comments and suggestions.

Yours sincerely,

Shi Liu, PhD
E-mail: shiliu2199@163.com